Author’s response to reviews

Title: Early and late mortality after malaria in young children in Papua, Indonesia

Authors:

Dewi Patriani (dewipatriani@yahoo.com)
Eggi Arguni (eggiarguni@gmail.com)
Saber Dini (saber.dini@unimelb.edu.au)
Paulus Sugianto (sugiantopaulus@gmail.com)
Afdhal Hasanuddin (afdalhs_805@yahoo.co.id)
Daniel Lampah (aditimika@yahoo.com)
Nicholas Douglas (nicholas.douglas@gmail.com)
Nicholas Anstey (Nicholas.Anstey@menzies.edu.au)
Julie Simpson (julieas@unimelb.edu.au)
Ric Price (rprice@menzies.edu.au)
Jeanne Rini Poespoprodjo (didot2266@yahoo.com)

Version: 4 Date: 19 Sep 2019

Author’s response to reviews:

Dear Dr. Oliver Cocks,

Thank you for the good news.

I have uploaded the database that was used for the analysis as a supplementary material.

Please find below my response to the requested revisions.

1. Please change the section heading “Introduction” to “Background”.

The heading has been changed to "Background".

2. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

I have clarified the role of the funding body in the main manuscript document.

3. The individual contributions of all authors to the manuscript should be specified in the Authors’
Contributions section. Currently the individual contributions of Nicholas M Anstey are not included. Guidance and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship.

I apologize for the mistake. I have now included Nicholas M Anstey in the Authors' Contribution section. Thank you.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

This is well noted and will do accordingly.

Thank you and Kind regards,

Rini